Author's response to reviews

Title: Prescribing cycle training intensity from the six-minute walk test for patients with COPD

Authors:

Muhammad R Zainuldin (mzai0920@mail.usyd.edu.au)
Danielle Knoke (dkno4772@mail.usyd.edu.au)
Martin G Mackey (m.mackey@usyd.edu.au)
Nia Luxton (n.luxton@usyd.edu.au)
Jennifer A Alison (j.alison@usyd.edu.au)

Version: 3 Date: 26 June 2007

Author's response to reviews:

Dear BMC Pulmonary Medicine,

The following revisions have been made:
1. Removal of the Registration number from the title
2. The inclusion of all authors in reference 6 and the removal of "et al"
3. The removal of the figure title from Figure 1
4. Slight reformatting of Figure 1 to make it a little more compact
5. Correction of email addresses for Nia Luxton and Jennifer Alison

I hope that you will find the manuscript ready for publication.

Best wishes,

Jennifer Alison